Author's response to reviews

Title: Interventions aimed at improving the nursing work environment: a systematic review

Authors:

Donna MJ Schalk (donna.schalk@han.nl)
Marloes LP Bijl (m.bijl@student.maastrichtuniversity.nl)
Greta G Cummings (gretac@ualberta.ca)
Ruud JG Halfens (r.halfens@zw.unimaas.nl)
Louk Hollands (l.hollands@zw.unimaas.nl)

Version: 2 Date: 26 August 2009

Author's response to reviews:

Dear Martin Eccles and Brian Mittman, Editors-in-Chief,

Revisions to our manuscript "Interventions Aimed at Improving the Nursing Work Environment: A Systematic Review" have been completed as required for publication in Implementation. Changes to the abstract, for example, now conform to the journal style and the amount of content is reduced to 350 words. Once again, thank you for your consideration in reviewing our research manuscript for publication.